Author's response to reviews

Title: Incidental renal cell carcinoma presenting in renal transplant recipient with autosomal dominant polycystic kidney disease: a case report

Authors:

Toshihiro Misumi (lupinn3sei@yahoo.co.jp)
Kentaro Ide (ideken@hiroshima-u.ac.jp)
Takashi Onoe (tonoe6060@yahoo.co.jp)
Masataka Banshodani (mban1127@yahoo.co.jp)
Hirofumi Tazawa (thiores@gmail.com)
Yoshifumi Teraoka (gc8teraoka@yahoo.co.jp)
Ryuichi Hotta (hotta.kuma@gmail.com)
Masahiro Yamashita (shanchan06291978@yahoo.co.jp)
Hirotaka Tashiro (htashiro@hiroshima-u.ac.jp)
Hideki Ohdan (hohdan@hiroshima-u.ac.jp)

Version: 4 Date: 27 March 2012

Author's response to reviews: see over
Dear Editor,

Thank you for sending the reviewers’ comments to us. We are grateful that you are willing to reconsider our manuscript entitled “Incidental renal cell carcinoma in an autosomal dominant polycystic kidney disease patient who underwent simultaneous bilateral native nephrectomy and living donor renal transplantation: a case report” (MS: 4374429966412482) for publication in *Journal of Medical Case Reports*. On the basis of your comments, we have changed the following title: “Incidental renal cell carcinoma presenting in renal transplant recipient with autosomal dominant polycystic kidney disease: a case report.”

Thank you once again for considering this revised manuscript for publication in *Journal of Medical Case Reports*.

Sincerely yours,

Kentaro Ide, M.D.
Department of Surgery, Division of Frontier Medical Science, Programs for Biomedical Research, Graduate School of Biomedical Sciences, Hiroshima University, 1-2-3 Kasumi Minami-ku Hiroshima 734-8551, Japan.
TEL: +81-82-257-5222, FAX: +81-82-257-5224,
E-mail: ideken@hiroshima-u.ac.jp